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• Experiments continue to run efficiently.
• Performed maintenance on near detector during 

downtime associated with tour.
« Electronics repairs (few channels)
« Work to improve magnet diagnostics

* Bdot system noise filters (will still need more work).
* Improved temperature readout.

« Installed remote reset for gateway computer.

• Cutover to faster network provider from Soudan.
« Hope to see benefits soon.
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• Target leak controlled by Helium backpressure.
« Increase late last week in He flow rate.

* No immediate concerns but increased caution.

• Smoke detector alarm in muon alcove #2 on Sunday 
afternoon.
« Very few flammables in that area.
« Controlled access by Fire Department detected no problems.
« Running with backup smoke detector.

• We have been very interested in the efforts being made 
to understand the optimal cycle times and timeline 
strategies for mixed mode operation. Thanks!
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Protons on target – two 
weeks.

Our estimate of npot – 4.3 x 1018
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00:00 to 00:00 CDT


